Author's response to reviews

Title: A yoga intervention for diabetes risk reduction: A pilot randomized controlled trial

Authors:

Kelly A McDermott (mcdermottk@ocim.ucsf.edu)
M. Raghavendra Rao (raghav.hcgrf@gmail.com)
Raghuram Nagarathna (rnagaratna@gmail.com)
Elizabeth J Murphy (emurphy@medsfgh.ucsf.edu)
Adam Burke (aburke@sfsu.edu)
R. Hongasandra Nagendra (hmr@vyasa.org)
Fredrick M Hecht (rhecht@php.ucsf.edu)

Version: 2 Date: 8 October 2013

Author's response to reviews: see over
Re: Publication of the manuscript “A yoga intervention for diabetes risk reduction: A pilot randomized controlled trial” (Revision to meet editorial requirements)

Dear Editor,

This version of the previously submitted manuscript includes the following requested editorial revisions:

1) Acknowledgement is included with funding details for the primary authors during the drafting of the manuscript as well as conduct of the study
2) We have reviewed the manuscript to ensure that it appropriately follows CONSORT guidelines
3) We have included the CONSORT checklist with the adaptation for nonpharmacological trials (NPT) as an additional file

BMC Complementary and Alternative Medicine (BMC CAM) has a reputation for publishing high quality CAM research and we believe that our manuscript, reporting results from a rigorous pilot study entitled “A yoga intervention for diabetes risk reduction: A pilot randomized controlled trial” would make an excellent contribution to the journal.

Thank you for your time and consideration of our manuscript.

Warmly,

Kelly McDermott, PhD
Osher Center for Integrative Medicine
University of California, San Francisco
1545 Divisadero St., #301
San Francisco, CA 94115